Author's response to reviews

Title: Varicella: epidemiological aspects and vaccination coverage in Veneto Region.

Authors:

Vincenzo Baldo (vincenzo.baldo@unipd.it)
Tatjana Baldovin (tatjana.baldovin@unipd.it)
Francesca Russo (francesca.russo@regione.veneto.it)
Marta C Busana (marta.busana@unipd.it)
Cinzia Piovesan (Cinzia.Piovesan@regione.veneto.it)
Greta Bordignon (greta_bordignon@yahoo.it)
Aurore Giliberti (auroregiliberti@gmail.com)
Renzo Trivello (renzo.trivello@unipd.it)

Version: 3 Date: 27 August 2009

Author's response to reviews:

MS: 3008541722602388

Varicella: epidemiological aspects and vaccination coverage in Veneto Region.

Vincenzo Baldo, Tatjana Baldovin, Francesca Russo, Marta C Busana, Cinzia Piovesan, Greta Bordignon, Aurore Giliberti and Renzo Trivello

We are resubmitting the above manuscript entitled “Varicella: epidemiological aspects and vaccination coverage in Veneto Region”. We appreciate your comments and review which improved the report substantially.

Enclosed you will find the revised version, including the sections Competing interest and Authors’ contributions.